Author's response to reviews

Title: Patterns of complementary and alternative medicine use amongst outpatients in Tokyo, Japan

Authors:

Satoshi Hori (satoshi@doctors.org.uk)
Iordan Mihaylov (imihaylov@doctors.org.uk)
Joana C. Vasconcelos (jv284@medschl.cam.ac.uk)
Malcolm McCoubrie (malc@audnmalc.demon.co.uk)

Version: 6 Date: 22 April 2008

Author's response to reviews: see over
Editor  
BMC Complementary and Alternative Medicine  

22\textsuperscript{nd} April 2008  

Dear Editor  

Re: MS: 1428371468179573 - Patterns of complementary and alternative medicine use amongst outpatients in Tokyo, Japan  

Thank you for your e-mail. We have made the requested formatting changes to our manuscript.  

Revisions to this manuscript  

1) References - The reference list should contain all authors names if there are 30 or below. The term ‘et al’ should not be used.  

Done  

2) Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.  

Done
We hope that the revised manuscript is now ready for publication.

Yours sincerely,

Dr Satoshi Hori
(on behalf of the authors)